Author's response to reviews

Title: Increased IR-A/IR-B Ratio in Non-small Cell Lung Cancers Associates with Lower Epithelial-Mesenchymal Transition Signature and Longer Survival in Squamous Cell Lung Carcinoma

Authors:

jiaqi huang (drijiangliyan@163.com)
wei zhu (drzhuwei@yeah.net)
katie streicher (streicherkatie@163.com)
chris morehouse (morehousechris@163.com)
phillip brohawn (broughnphilip@126.com)
xiaoxiao ge (gexiaoxiao1@163.com)
harvey dong (dongharvey1@163.com)
lucy yin (yinlucy@yeah.net)
mike zhu (zhumike@126.com)
yi gu (drguyi@163.com)
koustubh ranade (ranadekoustubh@163.com)
brandon w higgs (higgsbrandonw@163.com)
yihong yao (yaoy@medimmune.com)
jiaqi huang (huangj066@126.com)

Version: 3
Date: 15 January 2014

Author's response to reviews: see over
Dear Editor,

We would like to submit our manuscript entitled "Increased IR-A/IR-B Ratio in Non-small Cell Lung Cancers Associates with Lower Epithelial-Mesenchymal Transition Signature and Longer Survival in Squamous Cell Lung Carcinoma", which we wish to be considered for publication in BMC Cancer. We have revised our manuscript and added the sections of ‘Competing interests’ and ‘Competing interests’.

We confirm that the manuscript, or its contents in some other form, has not been published previously by any of the authors and/or is not under consideration for publication in another journal at the time of submission. All authors have reviewed the final version of the manuscript and approve it for publication. Thank you very much for your kind help and considering for potential publication.

Best wishes.

Sincerely yours,

Jiaqi Huang, Ph.D

15 January 2014